Author's response to reviews

Title: The Correlation Between Psychological Status and Oral Conditions with Self-perceived Halitosis among Jordanian Patients

Authors:

Firas Q Alzoubi (firasz@just.edu.jo)
Jumana A Karasneh (jumana2003@yahoo.com)
Nidal M Daamseh (dr.ndaamsa@hotmail.com)

Version: 2
Date: 23 February 2015

Author's response to reviews: see over
February 23, 2015

Managing Editor
BMC Oral Health

Dear Sir:

We have made the following modifications according to editorial requests as follow:

1) Authors should consider and write about the "publication of clinical datasets": In the study design and subjects section; the sentence was modified as follow; The aims and methodology of the study including possible future publication of clinical datasets were fully explained to all participants and a consent form was signed prior to interview and examination in compliance with the Declaration of Helsinki.

2) Large tables, table 1 and 3, do not fully appear in the manuscript. Authors should follow the submission guideline: Table 1 and 3 were uploaded separately as additional files.

Sincerely,

Firas Alzoubi, MD, FRCS
Chairman, Special Surgery Department
Consultant and Associate Professor of Otolaryngology
Faculty of Medicine
Jordan University of Science and Technology
PO Box 3030, 22110
Irbid Jordan
Email: firasz@just.edu.jo
Mobile: +962795411417